Author’s response to reviews

Title: Modelling the impact of improving screening and treatment of chronic hepatitis C virus infection on future hepatocellular carcinoma rates and liver-related mortality

Authors:

Matthew E Cramp (matthew.cramp@nhs.net)
William R Rosenberg (w.rosenberg@ucl.ac.uk)
Stephen D Ryder (stephen.ryder@nuh.nhs.uk)
Sarah Hindman (sarah.hindman@centerforda.com)
Julie Parkes (jules@soton.ac.uk)

Version: 2
Date: 28 March 2014

Author’s response to reviews:

Manuscript revised to clarify in the CI the role that the company, Gilead Sciences Ltd., had in study design and dataset selection.